\documentclass[11pt,reqno]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{xcolor}
\usepackage[colorlinks]{hyperref}
 

\usepackage{titlesec}
\titlespacing{\section}{0pt}{2.5ex}{1.5ex}
\titlespacing{\subsection}{0pt}{1.5ex}{1ex}
\titlespacing{\subsubsection}{0pt}{1.5ex}{1ex}
\titleformat{\section}{\large\bfseries\centering}{\thesection}{1em}{}
\titleformat{\subsection}[runin]{\bfseries}{\thesubsection.}{0.5em}{}[.\mbox{\ }]
\titleformat{\subsubsection}[runin]{\bfseries}{\thesubsubsection.}{0.4em}{}[.\mbox{\ }]





%%%%%---------- ORCiD
\newcommand\orcidicon[1]{\href{https://orcid.org/#1}{\includegraphics[scale=0.02]{orcid.pdf}}}


\def\udcs{511.1} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{11A51} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem*{theorem}{Main Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}



\hypersetup{linkcolor=blue,filecolor=black,urlcolor=blue, citecolor=blue}

\def\logo{\textcolor{blue}{{\bf\Huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}}

\def\semrtop
     {
  \vbox{
     \noindent\logo \hspace{-3mm}  \parbox{12cm}{     \begin{center}
     {\LARGE \textsf{СИБИРСКИЕ ЭЛЕКТРОННЫЕ}} \\[2mm]
     {\LARGE \textsf{МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ}} \\[2mm]
     {\Large \textsf{Siberian Electronic Mathematical Reports}} \\[1mm]
     {\Large\tt{\textcolor{blue}{http://semr.math.nsc.ru}}}\\[1mm]
     \raisebox{1ex}{ISSN 1813-3304 }
     \vspace*{-3mm}
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
}
     \vspace{-1mm}
     \noindent
      \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa!} \\
     \hline\hline
     \end{tabular}

     \vspace{3mm}
     {\flushleft\it Vol. 20, No. 2, pp. 144--145 (2023) \hspace{\fill}{\rm\small УДК \udcs \hspace{-4mm}}}
     \newline  %Volume, pages in Russian, filled by Editorial board
     	{\rm\small \textcolor{blue}{https://doi.org/10.33048/semi.2023.20.???}}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaa\;}{\rm\small MSC\ \ \mscs }
  }%\vbox
}




\begin{document}
\begin{thebibliography}{1} % Ссылки на литературу приводятся только на английском языке (даже если у источника нет англоязычной версии).

\bibitem {Simon} N. J. A. Sloane, S. Plouffe, \href{https://shop.elsevier.com/books/the-encyclopedia-of-integer-sequences/sloane/978-0-08-057186-7}{\it The Encyclopedia of Integer Sequences} , Academic Press, (1995),  \textbf{ISBN 9780125586306} (includes this sequence, page 319).

\bibitem {Connor} O'Connor, John J., Robertson, Edmund F., \href{https://mathshistory.st-andrews.ac.uk/Biographies/Al-Haytham/}{\textit{ \guillemotleft Abu Ali al-Hasan ibn al-Haytham\guillemotright}}, MacTutor History of Mathematics archive, University of St Andrews.

\bibitem {Edward} Edward Waring, \href{https://books.google.dz/books id=1MNbAAAAQAAJ&pg=PA380&redir_esc=y#v=onepage&q&f=false }{\textit{Meditationes Algebraicae}}, In the third edition of Waring's (1782), Wilson's theorem appears as problem 5 on page 380 (in Latin). 

\bibitem {Lagrange} J. L. Lagrange, \href{https://books.google.dz/books?id=_-U_AAAAYAAJ&pg=PA125&redir_esc=y#v=onepage&q&f=false}{\textit{ \guillemotleft D\'{e}monstration d'un th\'{e}or\`{e}me nouveau concernant les nombres premiers\guillemotright}} (Proof of a new theorem concerning prime numbers), Nouveaux M\'{e}moires de l'Acad\'{e}mie Royale des Sciences et Belles-Lettres (Berlin), vol. 2, pages 125-137 (1771).

\bibitem {Burton} Burton, David M., \href{https://jontalle.web.engr.illinois.edu/uploads/298/HistoryMath-Burton.85.pdf}{\textit{The History of Mathematics : An Introduction }} (7th ed, 2011), McGraw-Hill, \textbf{ISBN 9780073383156}, page 514 .

\bibitem {Legendre} A. M. Legendre, \href{https://gallica.bnf.fr/ark:/12148/bpt6k426107/f79.item#}{\textit{ Th\'{e}orie des nombres}}, Paris, (1830), page 10. 

\bibitem {Cassels} J.W.S. Cassels, A. Fr\"{o}hlich, \href{https://www.math.arizona.edu/~cais/scans/Cassels-Frohlich-Algebraic_Number_Theory.pdf}{\textit{Algebraic Number Theory}}  Academic Press,(1967), page 9.




\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}




\thispagestyle{empty}

\title[Werner Schulte's conjecture]{\large A simple proof of Werner Schulte's conjecture}


\author[Djamel HIMANE]{{\bf D. Himane}\protect\orcidicon{0000-0001-5956-8758}}



\address{Djamel Himane  % Контактные данные всех авторов и место работы указываются только на английском языке    Algiers, Algeria
\newline\hphantom{iii} LA3C Laboratory,  
\newline\hphantom{iii} Faculty of Mathematics, USTHB, 
\newline\hphantom{iii} Po. Box 32, El Alia, 16111, 
\newline\hphantom{iii} Bab Ezzouar, Algiers, Algeria.}%
\email{\textcolor{blue}{dhimane@usthb.dz)}}%

\thanks{\sc Himane, D., A simple proof of Werner Schulte's conjecture}
\thanks{\copyright \ 2024 Himane D}
%\thanks{\rm The work is supported by RFFI (grant 03-01-11111)}
\thanks{\it Received January, 1, 2024, Published December, 31, 2024}%


\semrtop \vspace{1cm}
\maketitle {\small

\vspace{-12pt}
\centerline{{\it Communicated by} {\sc *****}
%({\it filled by Editorial board})
}

\bigskip
\bigskip

\begin{quote}
\noindent{\bf Abstract:} Lately, Werner Schulte has conjectured that for all positive $n>1$, $n$ divides \: $\frac{(n-2)! \: (n-1)!}{2^{n-3}}  +  4 $ \: if and only if $ n $ is prime.     
 In this paper, We use elementary methods, to give a simple proof of this conjecture.   
 \medskip

\noindent{\bf Keywords:} Legendre's formula, p-adic valuation, Wilson Theorem, Fermat's Little Theorem.
 \end{quote}
}

\bigskip



\section{Introduction and statement of results } \: 
Taking  
\begin{equation}
    A_{n}\:=\:1\:(1+2)\:(1+2+3)\:...\:(1+2+3+...+(n-1)) 
\end{equation}
   the question is: does this sequence of positive integers significant? \\ In an attempt to find an explanation for this sequence. The triangular number is \quad $ T_{s}= 1+2+\cdots + s = \dfrac{s(s+1)}{2} $, as shown
   \begin{equation} \label{eq.1} 
   A_{n}\:=\: \prod_{j=1}^{n-1}\: T_{j}\:= \:\prod_{j=1}^{n-1}\: \frac{j(j+1)}{2}\:= \:\frac{(n-1)! \: n!}{2^{n-1}}  
\end{equation}

The  first terms of this sequence are $ 1,1,3,18,180,2700,56700, ...  \:  $   (sequence \href{https://oeis.org/A006472/internal}{A006472} in the OEIS \footnote{ \href{https://oeis.org/A006472/internal}{A006472}, The On-Line Encyclopedia of Integer Sequences.}), also (sequence M3052 \cite{Simon} ). 
 In the \href{https://oeis.org/A006472/internal}{OEIS}, we found a conjecture attributed to W. Schulte, concerning the previous sequence. for which we will present a simple proof.

  The main result we seek to prove is

\begin{theorem}[\emph{Werner Schulte's conjecture}] \:  \\
For all positive $n>1$, $n$ divides \: $\frac{(n-2)! \: (n-1)!}{2^{n-3}}  +  4 $ \: if and only if $ n $ is prime.
\end{theorem}

\section{Preliminaries}


For all prime numbers $ p $ and non-negative integer $ n $, let $ {\displaystyle v_{p}(n)} $ is the exponent of the greatest power of $ p $ dividing $ n $ (i.e. the \textit{p}-adic valuation of $ n $). Then 
\begin{equation} \label{eq.2} 
     {\displaystyle v_{p}(n!)=\sum _{i=1}^{\infty }\left\lfloor {\frac {n}{p^{i}}}\right\rfloor ,} 
\end{equation}

where $ {\displaystyle \lfloor x\rfloor } $  is the floor function. The formula of Legendre \eqref{eq.2} gives an expression to the greatest power of a prime $ p $ the exponent of which divide the factorial $ n! $ \cite{Legendre}. \:
Whereas the right-hand formula is an infinite sum, for every  particular value of $ n $ and $ p $, there are only a few nonzero terms: for each $ i $ sufficiently big as $ {\displaystyle p^{i}>n} $, we have  $ {\displaystyle \textstyle \left\lfloor {\frac {n}{p^{i}}}\right\rfloor = 0} $. \\
 We also have the recurrence relation: 
 \begin{equation}
   \displaystyle v_ {p} (n!) = \lfloor n / p \rfloor + v_ {p} (\lfloor n / p \rfloor!)   
 \end{equation}
Legendre's formula can also be reformulated in terms of a base-$ p $ expansion of $ n $. Let $ {\displaystyle s_{p}(n)} $ denotes the sum of the digits in the base-\emph{p} expansion of $ n $, then

\begin{equation}
   {\displaystyle v_{p}(n!)={\frac {n-s_{p}(n)}{p-1}}}.
\end{equation}

You can easily check the veracity of the following relationships. For any positive integers $ a, b $ and any prime $ p $ \cite{Cassels}. 
 
\begin{itemize}
    \item $ {\displaystyle \quad v_{p}\left(a \: b \right) = v_{p} (a) + v_{p}(b)} . $
    \item $ {\displaystyle \quad v_{p}\left({\frac {a}{b}}\right)= v_{p}(a)- v_{p}(b)} . $
    \item $ {\displaystyle \quad \text{if} \: a \mid b, \: then \quad v_{p}(a)\leq v_{p}(b)}   $, in particularly $  v_{p}(a) < a $.

\end{itemize} 

 The proof of the following lemma is known and easy, but is included here for completeness as the proof is short.
 
\begin{lemma} \label{Lemma 1} 
   For all positive integers $ m > 4 $, we have $ \: \:  m < 2^{m/2} $.   

\end{lemma}

 \begin{proof} 
 For the induction step, note that instead of using that the left side increases by $ 1 $, we will use the increase in multiplication (since the right side increases by a multiplicative factor) to get
  \begin{equation*}
    m   <   2^{m/2}  
     \end{equation*} 
     equivalent to    
 \begin{equation*}  
  m   \left(  \frac{m+1}{m} \right)     \: < \: 2^{m/2}   \left(\frac{m+1}{m}\right)  
   \end{equation*}
    equivalent to  
    \begin{equation} \label{eq.3} 
    m +1 \: < \: 2^{m/2} \: \left(\frac{m+1}{m}\right).    
 \end{equation}


Next, let
\begin{equation*}
   \mathrm{g}(w) = \sqrt{2}-\frac{w+1}{w} = (\sqrt{2}-1)- \frac{1}{w}.  
\end{equation*}

This gives $ \mathrm{g}(5)= (\sqrt{2} - 1)- \frac{1}{5} \approx 0.2142 $. Also, $ \mathrm{g}'(w) = \frac{1}{w^{2}} > 0 $, which means $ \mathrm{g}(w) $ is a strictly increasing function for $ w\geq 5 $, so it's always positive then,  
 \begin{equation*}
     \sqrt{2} - \frac{w+1}{w} \:> \: 0 \: \quad \: \quad implies \: \quad \: \quad \sqrt{2}  \: > \: \frac{w+1}{w} 
\end{equation*}
   Thus, for $ m\geq 5 $,  
\begin{equation*}
     \frac{m+1}{m} \quad < \quad 2^{1/2}  \: \quad \quad \: gives \: \quad \: \quad 2^{m/2} \: (\frac{m+1}{m}) \quad  < \quad  2^{(m+1)/2} \quad     
\end{equation*}


Using this in \eqref{eq.3}  gives
\begin{equation*}
   m+1 \quad < \quad  2^{(m+1)/2}  
\end{equation*}


which completes the induction step.
As a result for $ m > 4 $, we get  
\begin{equation*}
   m \: < \: 2^{m/2}    \: \quad \: \quad \quad \: implies   \quad \quad \: \quad  v_{2} (m) \: < \: \frac{m}{2}.
\end{equation*}

\end{proof}


\begin{lemma}[{\small Wilson Theorem}]\label{Lemma 2}If $p$ is prime then {\small $(p-1)!+1$} is divisible by $p$. \end{lemma}
  This theorem was affirmed by Alhazen (c. 965-1000 A.D.), \cite{Connor} and, in the eighteenth century, from J. Wilson \cite{Edward}. Proof of this came from J. L. Lagrange in 1771 \cite{Lagrange}.  An integer primality test derives from the Wilson theorem: a non-negative integer $ n > 1 $ is prime, If and only if
\begin{equation}
 (n-1)!+1 \: \equiv 0 \:  \pmod n. 
\end{equation}


\begin{lemma}[Fermat's Little Theorem]  \label{Lemma 3}
 For any positive integers $ a $ which cannot be divided by a prime number $p$, we have 
\begin{equation}
  a^{p-1} \equiv 1 \:  \pmod p.   
\end{equation} 
  \end{lemma}
 Among one of the basic results of the theory of elemental numbers Little theorem of Fermat, This is the basis of Fermat's primality test. this theorem is called from Pierre de Fermat, who declared during 1640. It is referred to as the:  \guillemotleft Little theorem\guillemotright \, to set it apart from Last Theorem of Fermat.  (see page 514,  \cite{Burton} ) .






\section{Proof of the Main Theorem}
\begin{proof}  
 \textbf{First step.}  It is shown that if $n$ is prime, then $n$ divides
 \begin{equation} 
 \frac{(n-2)!(n-1)!}{2^{n-3}} + 4 . 
 \end{equation}
We assume $ n=p $ is odd prime, thus from Wilson's theorem \ref{Lemma 2}, we get $ p $ divided both $ (p-1)!+1 $ and $ (p-2)!-1 $. 
  Hence  $ p $ divided $ (p-1)!(p-2)!+1 $. \\ 
Instead,  from little theorem of Fermat \ref{Lemma 3}, we get $ p $ divided $ 2^{p-1} - 1 $, hence from \eqref{eq.1}
 we get
 \begin{equation*}
 2 \times A_{p-1} \: = \: \frac{4 \:(p-2)! \: (p-1)!}{2^{p-1}} \: \equiv 4 \:\:(p-2)! \: (p-1)! \: \equiv -4   \pmod p.
  \end{equation*}
  So $ p $ divides the number  
  \begin{equation*}
  \frac{(p-2)! \: (p-1)!}{2^{p-3}}  +  4 .   
 \end{equation*}
   Finally, for $n = 2 $, the validity of the property can be easily checked.
  \\
  \quad \textbf{Second step.} \: We show that if $ n $ is a composite, then
    \begin{equation*}
  \frac{(n-2)! \: (n-1)!}{2^{n-3}}  +  4  
    \end{equation*}
  It is not divisible by $ n $. \\
 We assume $ n = p q $ is a composite, where $ 1 < p \leq q < n $ three sub-cases arise : \\
 \begin{itemize}
\item \textbf{First case.} if $ n = 4 $ , then $ \frac{(n-2)! \: (n-1)!}{2^{n-3}}  +  4 \: = \: 10 $, It is not divisible by $ n $. 
\\
\item \textbf{Second case.} if $ n $ odd $ (n>4) $, then $ n\mid (n-2)!(n-1)! $ and $ \gcd(n, 2^{n-3})=1$ from Euclid's lemma, we have $ n\mid \frac{(n-2)! \: (n-1)!}{2^{n-3}} $, Hence  
 $ \frac{(n-2)! \: (n-1)!}{2^{n-3}}  +  4 \: \equiv \: 4 \pmod n, $ So $\frac{(n-2)! \: (n-1)!}{2^{n-3}}  +  4 $ it is not divisible by $ n $. \\
\item \textbf{Third case.} if $ n $ even $ (n=2^{v_{2}(n)}\times k > 4)$ where $k$ odd, then $ n\mid (n-2)!(n-1)! $ from Euclid's lemma, we get $ k\mid 2 \times A_{n-1} $
 , hence  $ n \mid 2 \times A_{n-1} $ if and only if \[ v_{2}(n)\leq v_{2}(2 \times A_{n-1}). \]
 Through the above properties of \emph{p}-adic valuation, we get 
 \[  v_{2}(2 \times A_{n-1})= v_{2}((n-1)!)+ v_{2}((n-2)!)- v_{2}(2^{n-3})  \]
 We also have the recurrence relation:
   \begin{equation*}
 \displaystyle v_ {2} ((n-1)!) = v_ {2} (n-1) + v_ {2} ((n-2)!) = v_ {2} ((n-2)!),  
    \end{equation*}
 Because $ n-1 $ odd, and
    \begin{equation*}
  \displaystyle v_ {2} ((n-2)!) = \lfloor \frac{n-2}{2} \rfloor + \lfloor \frac{n-2}{4} \rfloor + v_ {2} (\lfloor \frac{n-2}{4} \rfloor!) \geq \frac{n-2}{2} + \frac{n-2}{4} 
     \end{equation*}
Therefore 
    \begin{equation*}
  v_{2}\left(2 \times A_{n-1} \right)\geq \left( \frac{n-2}{2} + \frac{n-2}{4}\right)  + \left( \frac{n-2}{2} + \frac{n-2}{4}\right)  - \left(n-3 \right)  
     \end{equation*}
As a result  
   \begin{equation*}
  v_{2}(2 \times A_{n-1}) \: \geq \:  \frac{n}{2} .  
     \end{equation*}
From Lemma \ref{Lemma 1}, we obtain
\begin{equation*}
   \frac{n}{2} \: >   \: v_ {2} (n) .  
     \end{equation*} 
     Then 
     \begin{equation*}
  v_{2}(2 \times A_{n-1}) \:  \geq \: v_ {2} (n) .  
     \end{equation*}
Hence
     \begin{equation*}
 \frac{(n-2)! \: (n-1)!}{2^{n-3}}  +  4 \: \equiv \: 4 \pmod n,   
    \end{equation*}
  So
   \begin{equation*}
  \frac{(n-2)! \: (n-1)!}{2^{n-3}}  +  4
  \end{equation*}
      is not divisible by $ n $.  
 \end{itemize}
In summary, we have obtained to all positive integer $ n>1 $, that  $n$ divides 
\begin{equation*}
\frac{(n-2)! \: (n-1)!}{2^{n-3}}  +  4
\end{equation*}
   if and only if $ n $ is prime.
\end{proof}
\bigskip

\end{document}
